Author's response to reviews

Title: Bloodstream infection following 217 consecutive systemic-enteric drained pancreas transplants

Authors:

Natalie R Berger (Natalie.Berger@uibk.ac.at)
Siegmund Guggenbichler (Siegmund.Guggenbichler@uibk.ac.at)
Wolfgang Steurer (W.Steurer@vodafon.de)
Christian Margreiter (Christian.Margreiter@uibk.ac.at)
Gert Mayer (Gerd.Mayer@uibk.ac.at)
Reinhold Kafka (Reinhold.Kafka@uklibk.ac.at)
Walter Mark (Walter.Mark@uibk.ac.at)
Alexander R Rosenkranz (Alexander.Rosenkranz@uibk.ac.at)
Raimund Margreiter (Raimund.Margreiter@uibk.ac.at)
Hugo Bonatti (Hugo.bonatti@uklibk.ac.at)

Version: 2 Date: 2 February 2006

Author's response to reviews:

we have tried to address all valuable comments, which have been brought up by the reviewers. They have been added to the revised manuscript in BOLD.
This includes the comments on the management and prevention of IAI, the impact of UTI and the importance to remove indwelling devices as well as the use of antifungal agents to prevent fungemia. The text was edited for appropriate English.
We hope the revisions are satisfactory and the article can be published.